

Author's response to reviews

Title: A high confidence, manually validated human blood plasma protein reference set

Authors:

Susann Schenk (sschenk@bmb.sdu.dk)
Gary J Schoenhals (gary@bmb.sdu.dk)
Gustavo de Souza (gso041@uib.no)
Matthias Mann (mmann@biochem.mpg.de)

Version: 6 **Date:** 14 August 2008

Author's response to reviews:

Dear Deepali Singhal,

RE:

MS: 1691558822166885

A high confidence, manually validated human blood plasma protein reference set. Susann Schenk, Gary J Schoenhals, Gustavo de Souza and Matthias Mann

We have completed the final revisions to our manuscript as instructed by BMC Medical Genomics. The revisions include the repair of the corrupted references, cropping of the figures, and renaming of the additional files so that their names reflect the order of their appearance in the manuscript text. We ask that you please upload the manuscript for us as we do not have access to do so ourselves. Please find the following files attached:

Manuscript:

0_edit_.doc

Figures:

1_cropped.pdf

2_cropped.pdf

3_cropped.pdf

4_cropped.pdf

5_cropped.pdf

6_cropped.pdf

7_cropped.pdf

8_cropped.pdf

9_cropped.pdf

10_cropped.pdf

Additional files:

(Please note that the names of these additional files has changed)

Additional_file_1.xls

Additional_file_2.doc

EndNote (version 10) library for references: blood_plasma_final.enl

Canvas X and Adobe Photoshop version 7 were used to manipulate the figures.

We trust that you will find our revisions to be satisfactory. We are looking forward to seeing our manuscript published in BMC Medical Genomics.

Thank you for your assistance.

Kind regards,

Gary Schoenhals and Susann Schenk